Author's response to reviews

Title: Modification of additive effect between vitamins and ETS on childhood asthma risk according to GSTP1 polymorphism: A cross-sectional study

Authors:

So-Yeon Lee (imipenem@hanmail.net)
Bong-Seong Kim (bskpded@hanmail.net)
Sung-Ok Kwon (jaja0609@khidi.or.kr)
Se-Young Oh (seyoung@khu.ac.kr)
Hye Lim Shin (ysb1126@hanmail.net)
Young Ho Jung (youngdr1@naver.com)
Eun Lee (unelee@hanmail.net)
Song-I Yang (qjit222@naver.com)
Hyung Young Kim (hawmok77@hanmail.net)
Ju-Hee Seo (juheesea@hanmail.net)
Byoung-Ju Kim (drkbi@hanmail.net)
Hyo-Bin Kim (hbkim1126@naver.com)
Ji-Won Kwon (pedas@naver.com)
Hae-Ran Lee (driran@hallym.or.kr)
Soo-Jong Hong (sjhong@amc.seoul.kr)

Version: 5
Date: 20 July 2015

Author's response to reviews: see over
Modification of additive effect between vitamin and ETS on childhood asthma risk by GSTP1 polymorphism: a cross sectional study

: Thank you for taking the time out of referee’s busy schedule to write to us. Your feedback regarding my revised manuscript is greatly appreciated and has assisted us in improving it. I have made some corrections and clarifications in the manuscript after going over editorial request.

The detail changes and reasons are summarized as below:

EDITORIAL REQUEST:

1)Figures: You have uploaded your figures twice in the final draft. Please remove the second copy. Also you have uploaded figures 1a and 1b separately- our system automatically numbers them as they are uploaded, so they now appear as fig 1 and fig 2 and as such don’t follow the figure legend. See http://www.biomedcentral.com/authors/figures for the appropriate way to configure the figures.

: Thank you for your kind comment. As your comment, we changed upload of our figures.

2)STROBE: As it is a cross-sectional study, adherence to STROBE guidelines is required. We request you upload the STROBE checklist and fill it in accordingly. You can find the checklist here http://www.equator-network.org/reporting-guidelines/strobe/

: We filled the STROBE checklist and upload it.

3)Title Page: All of the co-authors email addresses are to be listed on the title page according to BMC guidelines.

: We added co-authors email addresses on the title page

4)Authors Contribution: This section should be placed directly after the Competing Interests Section and should adhere to BMC guidelines- which can be found here http://www.biomedcentral.com/bmcpublichealth/authors/instructions/researcharticle

: We corrected authors contribution.

5)Proof reading by a native English speaker. There are still a few mistakes littered throughout the manuscript. For e.g. Table 1e: “Children include in this study (N=1111)” needs to read “included”

: This manuscript was received proofreading again by a native English speaker.